Author's response to reviews

Title: Impact of valvular heart disease on daily living activity of nonagenarians: The Leiden 85-plus study

Authors:

Thomas van Bemmel (t.van_bemmel@lumc.nl)
Victoria Delgado (v.delgado@lumc.nl)
Jeroen J Bax (j.j.bax@lumc.nl)
Jacobijn Gussekloo (j.gussekloo@lumc.nl)
Gerard J Blauw (g.j.blauw@lumc.nl)
Rudi G Westendorp (r.g.j.westendorp@lumc.nl)
Eduard R Holman (e.r.holman@lumc.nl)

Version: 4 Date: 29 March 2010

Author's response to reviews: see over
Dr. Melissa Norton  
Editor-in-Chief  
*BMC Geriatrics*

Subject: revision of the manuscript: 5109993402713548

Dear Dr. Norton,

We are delighted that our manuscript has been accepted for publication. As requested we have added extra context/background information in the background section of the abstract: “Data on the prevalence of valvular heart disease in very old individuals are scarce and based mostly on in-hospital series. In addition, the potential detrimental effect of valvular heart disease on the activities of daily living is unknown.”

All requested changes by the editorial production team have been performed.

Sincerely yours,

Thomas van Bemmel